Author’s response to reviews

Title: Molecular diagnosis in recessive pediatric neurogenetic disease can help reduce disease recurrence in families

Authors:

Mahmoud Issa (myissa2002@hotmail.com)
Zinayida Schlachetzki (z.schlachetzki@gmail.com)
Valentina Stanley (vstanley@ucsd.edu)
Renee George (reneegeorge@gmail.com)
Jennifer McEvoy-Venneri (jmcevoyvenneri@ucsd.edu)
Denice Belandres (dbelandres@ucsd.edu)
Hasnaa Elbendary (hasnaa_mohamed24@yahoo.com)
Khaled Gaber (krgaber@gmail.com)
Ahmed Nabil (ahmadnabil248@yahoo.com)
Mohamed Abdel-Hamid (mohamadnrc@hotmail.com)
Maha Zaki (dr_mahazaki@yahoo.com)
Joseph Gleeson (jogleeson@ucsd.edu)

Version: 4 Date: 19 Apr 2020

Author’s response to reviews:

April 19, 2020

Dear Editor,

Thank you for your feedback to improve our manuscript. We received the following comment from your office:
"Thank you for your response. However, your statement remains unclear. We kindly request that, in your “Consent for publication” section, you clarify whether written informed consent for publication of personal details, clinical details and/or clinical images was obtained from all of the participants. In the case of minors, which refers to individuals younger than the age of 18, consent for publication must be obtained from their parents or legal guardians. As such, in your “Consent for publication” section, please clarify whether written informed consent for publication of personal details, clinical details and/or clinical images was obtained from the parents or legal guardians of any participant under the age of 18.”
In response, we have modified the Consent for publication to the following:

"Written consent for publication of identifying images and other personal and clinical details of participants that compromise anonymity was documented in the consent form. Written consent for publication of identifying images or other personal or clinical details of participants under the age of 18 that compromise anonymity was obtained from parents or legal guardians and documented in the consent form."

We hope that this text is clearer, and that it removes any lingering ambiguity about consent for publication.

Many thanks,

Prof. Joseph Gleeson
Univ California San Diego